Author’s response to reviews

Title: Plectranthus amboinicus essential oil and carvacrol bioactive against planktonic and biofilm of oxacillin- and vancomycin-resistant Staphylococcus aureus

Authors:
Renata Costa (renata.albuq@gmail.com)
Sara Edwirgens Costa Benício Vasconcelos (sara.benicio@hotmail.com)
Hider Machado Melo (hidermbio@gmail.com)
Theodora Thays Arruda Cavalcante (theodorathays@gmail.com)
Francisco Eduardo Aragão Catunda Júnior (cearajr@gmail.com)
Mário Geraldo de Carvalho (mgeralduo@ufrrj.br)
Francisca Gleire Rodrigues Menezes (gleierodrigues@yahoo.com.br)
Oscarina Viana de Sousa (oscarinasousa@yahoo.com.br)

Version: 2 Date: 06 Sep 2017

Author’s response to reviews:
Dear Editor,

All suggested modifications were made.

Best regards,
Dr. Renata A. Costa

1. Please include the email addresses of all manuscript authors on your Title Page.
   All emails have been entered.

2. Thank you for including a Competing Interests section in your manuscript. However, we would ask you to please revise the statement which appears in this section. If you do not have any competing interests then, rather that 'not applicable', we would ask you to please state 'The authors declare that they have no competing interests' in this section.
The modification was made.

3. Thank you for uploading your figures into our online submission system. However, we would ask you to please upload each figure as a separate and individual 'figure' file. Please also note that figure legends should not be uploaded with these figures. The legends should instead be included as a list at the end of your manuscript document.

The figures were inserted individually.